Córdoba, April 7, 2015

Editors
BMC Ophthalmology

Dear Editors:

I am pleased to send to you our paper Systemic amyloidosis with bilateral conjunctival involvement: a case report to be considered for publication in your prestigious journal.

Conjunctival amyloidosis is a very rare disease, and in most cases it is unilateral and isolated without systemic amyloidosis.

In this paper we present a very infrequent case of a patient with bilateral conjunctival amyloidosis associated with asymptomatic systemic amyloidosis. We discuss the clinical picture, workup for diagnosis and treatment. As conjunctival amyloidosis presents generally as a chronic conjunctivitis, diagnosis is usually overlooked, as it may occur with systemic amyloidosis that may become symptomatic in advanced and severe cases, compromising the patient’s life.

All authors, and in particular the corresponding author, had full access to all the data in the study and take responsibility for the integrity of the data and the accuracy of the information presented.

We hope that you will find our paper suitable to be published in BMC Ophthalmology as a case report.
Yours sincerely,

Julio A. Urrets-Zavalía, M.D.
Corresponding author